Author's response to reviews

Title: Carney complex: Case report and review

Authors:

Shirish S Borkar (drshirishborkar@hotmail.com)
Sevagur Ganesh Kamath (sevagur@hotmail.com)
Nitin Kashyap (nitinkashyap1@yahoo.com)
Sunil CV Sagar (dmutyalasunil@yahoo.co.in)
Lakshmi Rao (lakupindi@hotmail.com)
Raj Warrier (rwarri@yahoo.com)
Aman Chauhan (dramanchauhan@hotmail.com)

Version: 3 Date: 15 February 2011

Author's response to reviews:

Respected sir I have done the following correction.

1. Now the authors in submission system and manuscript are in order and correct style.
2. References are now in square brackets.
3. Figures are individually uploaded.
4. Keywords removed.
5. Country name is written in the affiliation.